Author's response to reviews

Title: Adherence to physiotherapy clinical guideline acute ankle injury and determinants of adherence: a cohort study

Authors:

Philip J van der Wees (philip.vanderwees@epid.unimaas.nl)
Erik JM Hendriks (erik.hendriks@epid.unimaas.nl)
Mariette J Jansen (mjjansen@hetnet.nl)
Hans van Beers (h.vanbeers@student.unimaas.nl)
Rob A de Bie (ra.debie@epid.unimaas.nl)
Joost Dekker (j.dekker@vumc.nl)

Version: 4 Date: 21 May 2007

Author's response to reviews:

Dear Editor Production Team,

I have made the following formatting changes as requested:

1. Box 1 has been replaced by Table 1. The other tables are correspondingly renamed as Table 2-5. References in the text to the tables have been changed according to the new table numbers.

2. The author names in the submission system have been changed according to the names in the manuscript. The following surnames have been changed: van der Wees, van Beers, de Bie.

Sincerely,

Philip van der Wees